Author's response to reviews

Title: Improvement of endocytoscopic findings after per oral endoscopic myotomy (POEM) in esophageal achalasia; Does POEM reduce carcinogenesis in achalasia patients?

Authors:

Hitomi Minami (le7novembre@gmail.com)
Hajime Isomoto (hajimei2002@yahoo.co.jp)
Naoyuki Yamaguchi (naoyuki3334@hotmail.com)
Kayoko Matsushima (kmatsu@nagasaki-u.ac.jp)
Yuko Akazawa (akazaway@nagasaki-u.ac.jp)
Ken Ohnita (k-ohnita@nagasaki-u.ac.jp)
Fuminao Takeshima (ftake@net.nagasaki-u.ac.jp)
Toshiyuki Nakayama (toshi-n@nagasaki-u.ac.jp)
Tomayoshi Hayashi (toma@nagasaki-u.ac.jp)
Haruhiro Inoue (haruinoue777@gmail.com)
Kazuhiko Nakao (kazuhiko@nagasaki-u.ac.jp)

Version: 2 Date: 9 October 2012

Author's response to reviews: see over
Editor in Chief,
BMC Gastroenterology

Dear Professor Melissa Norton

On behalf of all the authors, I would like to ask you to consider our manuscript entitled “Improvement of endocytoscopic findings after successful Per oral endoscopic myotomy (POEM) in esophageal achalasia; Does POEM reduce carcinogenesis in achalasia patients?” for publication in “BMC gastroenterology”.

We would like you to consider this manuscript for publishing.

The manuscript has not been published previously, in any language, in whole or in part, and is not currently under consideration elsewhere. All authors have contributed to and agreed on the content of the manuscript, and the respective roles of each author. No financial support for this study was provided by any company or foundation. The authors report no conflicts of interest. All samples were obtained with the written informed consent of the patients prior to their inclusion, in accordance with the Helsinki Declaration and under approval by the Nagasaki University Hospital Ethics Committee.

Sincerely,

Hitomi Minami, MD
TEL +81-90-7450-4583
E-mail le7novembre@gmail.com
Department of Gastroenterology and Hepatology, Nagasaki University Hospital
Address 852-8501, 1-7-1, Sakamoto, Nagasaki, Japan

Corresponding author
Hajime Isomoto, MD, PhD
E-mail hajimei2002@yahoo.co.jp
TEL  +81-95-819-7481
FAX  +81-95-819-7482
Department of Gastroenterology and Hepatology, Nagasaki University Hospital
Address  852-8501, 1-7-1, Sakamoto, Nagasaki, Japan